\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\AtBeginDocument{\let\citet\cite\let\citep\cite}
\usepackage[numbers,sort&compress,numbers,sort&compress]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\let\@eISSN\@empty
\let\@pISSN\@empty
\def\eISSN#1{\gdef\@eISSN{#1}}
\def\pISSN#1{\gdef\@pISSN{#1}}
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\usepackage{qrcode}

\def\abbrvJournalTitle{Int. J. of Clin. Pharm. Med. Sci.}
\def\jshort{ijcpms}

\def\title#1{%
  \gdef\@title{%
%      \vspace*{-20pt}%
       \vspace*{-48pt}
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%
         \parbox{.1\linewidth}{\includegraphics[width=4.2cm]{pharmasprings_logo.png}}\hfill
%         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.729\linewidth}{%
          \begin{center}
          \vskip-4pt 
          \scshape\fontsize{18}{20}\selectfont\journalTitle\hfill\end{center}%
          \vspace*{-16pt}{\vspace*{-3.1pt}\fontsize{8}{9}\selectfont Published by Pharma Springs Publication\hfill  Journal Home Page: \href{https://pharmasprings.com/ijcpms/}{\color{blue}\underline{\smash{https://pharmasprings.com/ijcpms/}}}}\hspace*{4pt}\mbox{}}}%
       \par\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}


\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}


                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle\ \@copyrightYear;\space\ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    :\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[LO]{\fontsize{9.12}{11}\selectfont\textcopyright\ Pharma Springs Publication\ $|$\space \journalTitle}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
\def\bibjtitle#1{#1}
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}
\lefthyphenmin = 3
\makeatother
\date{}
\makeatletter
\usepackage{mdframed}
%\usepackage{balance}
\definecolor{unitednationsblue}{RGB}{222,234,246}
\definecolor{cpboxcolor}{RGB}{211,211,211}
\definecolor{citeboxcolor}{RGB}{242,242,242}
\def\printBio{%
  %\noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
  \noindent\begin{mdframed}[innerbottommargin=7pt,backgroundcolor=unitednationsblue,hidealllines=true,frametitle={ABOUT AUTHORS},innerrightmargin=20pt,rightmargin=4,innerleftmargin=7pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
   \@tempcnta=0
    \loop
      \advance \@tempcnta by 1
      \def\aubioCnt{\the\@tempcnta}
      \vskip1pt
%      \noindent\hspace{-2pt}%\fbox{
%     \vspace*{1pt}
      \raisebox{62pt}%{\begin{minipage}[t]{.35\linewidth}%
      \expandafter\ifx\csname authorImage\aubioCnt\endcsname\relax%
       \else%
         \includegraphics[height=60pt,width=60pt]{\csname authorImage\aubioCnt\endcsname}%height=145pt
       \fi%
      %\end{minipage}
      %}%}
      \hskip 4pt%\fbox{
      \raisebox{52pt}{\begin{minipage}[t]{.74\linewidth}%
        {\fontsize{8}{10}\selectfont{\csname authorName\aubioCnt\endcsname\\[5pt]}}
        \fontsize{8}{10}\selectfont\csname authorDetails\aubioCnt\endcsname\\
      \end{minipage}}
%      }
%     \vskip10pt%
    \ifnum\@tempcnta < \theaubio
    \repeat
%     \vspace*{-16.5pt}%
  \end{mdframed}
}
\usepackage{balance}

\makeatother


\def\citeauthor{~Akhila~saila~sree~T,
              ~Samyuktha~K,
              ~Nandini~Pandey,
              ~Bhargav~reddy~C}
\def\citetitle{A Case Report on Coarctation of the Aorta}
\def\copyrightstmt{This is an open access article distributed under the terms of the Creative Commons Attribution-NonCommercial- ShareAlike 4.0 License, which allows others to remix, tweak, and build upon the work non-commercially, as long as the author is credited and the new creations are licensed under the identical terms.}
\publishedDate{}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Feature enabled:
%citation-style: unsrt
%citation-type: numbered
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage{float}

\begin{document}


\def\RunningAuthor{Akhila Saila Sree T et al.}
\firstPage{14}
\articleType{Case Report}
\receivedDate{08 Feb 2021}
\acceptedDate{17 Feb 2021}
\revisedDate{15 Feb 2021}
\journalVolume{1}
\journalIssue{1}
\copyrightYear{2021}
\eISSN{2583-0953}
\journalDoi{10.26452/ijcpms.v1i1.182}
\def\authorCount{4}
\def\affCount{2}

\def\journalTitle{International Journal of Clinical Pharmacokinetics and Medical Sciences}

\title{A Case Report on Coarctation of the Aorta}
\author{Akhila~saila~sree~T\textsuperscript{*}\textsuperscript{1},  
            Samyuktha~K\textsuperscript{2},  
            Nandini~Pandey\textsuperscript{2}, 
            Bhargav~reddy~C\textsuperscript{2}~\\[5pt]{\textsuperscript{1}Santhiram College of Pharmacy, Nandyal, Kurnool Dist, Andhra Pradesh, India}~\\{\textsuperscript{2}Santhiram Medical College and General Hospital, Nandyal, Kurnool Dist, Andhra Pradesh, India}}

\begin{abstract}
Coarctation of the aorta is described as an innate cardiac anomaly comprising the narrow aortic section consisting medial stiffen with a bit in folding of the media and fortuitous neo tissue layer membrane (intimal). Coarctation of the aorta is the 6th most familiar lesion in hereditary/innate heart disease. Here, the localized constriction forms shelf like formation with an aberrant aperture or membranous curtain like formation with a central or aberrant aperture, though in spite of consolation of the anatomical restraint, the succeeding risk of premature morbidity and death keep on. The present study figures the ideal guidance of a disease from neonatal to adult life. It also includes the treatment of coarctation of the aorta by the evolution of transcatheter treatment for either the native and peculiarly repetitive coarctation of the aorta. Late obstacles, even after the proper fortunate treatment, are not common. So, lifelong follow-up is more vital in coarctation of aorta patients. 
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Coarctation of the Aorta,\newline Congenital Heart Disease,\newline Transcatheter Therapy,\newline Aberrant Aperture
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright{\Large$^\ast$}Corresponding Author\par Name:\ Akhila~saila~sree~T~\\ Phone:\ 8008609502~\\  Email:\ akhilasailasree@gmail.com
        \par}\vspace*{-13pt}\hrulefill~\\{\fontsize{12}{14}\selectfont\ifx\@eISSN\@empty\else eISSN: \@eISSN\fi\ifx\@pISSN\@empty\else\hfill pISSN: \@pISSN\fi}\par% 
        \textsc{DOI:}\ \href{https://doi.org/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/\@journalDoi}}}}\par%
        \vspace*{-12pt}\hrulefill\\[5pt]\hypersetup{allcolors=black}\qrcode[version=4,height=1.5cm]{https://doi.org/\@journalDoi}\hypersetup{allcolors=blue} \hspace*{10.5pt}\parbox[t]{5cm}{\raisebox{19pt}{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\\[-13pt]{{\fontsize{12}{14}\selectfont\raggedright Pharmasprings.com \\[-1pt]{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\\[-8pt]\hspace*{-1pt}\rule{.762\columnwidth}{1.2pt}}}}\hypersetup{allcolors=blue}%{\fontsize{12}{14}\selectfont Pharmasprings.com}\par%
        \makeatother
        
    
\section{Introduction}
The word coarctation referred as constrict/tighten and is originated from the latin coatare, i.e., to tightening inturn the stemming from co-and arctare and to repair vigorously. From artus, close, tight-coarctation of aorta (Co A) is a relatively common innate cardiac defect. In Coarctation of the aorta, a portion of the artery constricts, usually in nearer to the ductus arteriosus or the ligamentum arteriosum following the blockage. But the narrow area of the aorta may differ in location and also in the structure, wideness and length. Its incidence ranges from 5 - 8\% of all the innate cardiac defects. This state is most frequently diagnosed because of heart sounds or HTN found on general auscultation. Aortic blockage may be restored by the surgery or by the transcatheter operating procedures, the later it mainly go through the balloon angioplasty and stent installation. In back days, surgery has been widely used, but due to the death rate and extremely obstacles correlated with surgery, catheter operating procedure are exclusively used for the therapeutic outcome \unskip~\cite{1020278:21539464}. Balloon angioplasty in infants and stents in teens are fetching early choice of treatment for the management of coarctation of the aorta \unskip~\cite{1020278:21539468}. Classification of coarctation of aorta includes the: 


  
  \begin{enumerate}
  \item \relax Native coarctation of the aorta
  \item \relax Recurrent coarctation of the aorta
  \end{enumerate}
  It also includes preductus, ductus and post ductus coarctation of the aorta. Native coarctation of the aorta defines the detached tapering of the subsiding aorta that ensures from ridge-like stiffen of the media of aortic wall that projects into lumen facing the insertion of the ductus arteriosus. The beginning of the subclavian artery can be intricate with post-stenotic expansion of the aorta commonly encountered.

Recoarctation of the aorta is defined as restenosis succeeding of an inceptive triumphant surgical or


\begin{table*}[!htbp]
\caption{\boldmath {Clinical laboratory data} }
\label{tw-9b2802a9c87e}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.4528\linewidth-2\tabcolsep}p{\dimexpr.5472\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Tests & Test Value\\
\tblmidrule 
Hb &
  15.2g/dl\\
WBC Count &
  7800num/\ensuremath{\mu}l\\
Neutrophils &
  57\%\\
Eosinophils &
  3\%\\
Lymphocytes &
  33\%\\
Platelets count &
  2.92lakh/cumm\\
Serum uric acid &
  6mg/dl\\
Serum creatinine &
  1.04mg/dl\\
Serum urea &
  26mg/dl\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}
catheter deploys restoration and is thought to be peripheral to both a residual blockage or evolution of restenosis. This prevalence of recoarctation following the surgery is around 10\%and happens independently of the type of surgical restoration used \unskip~\cite{1020278:21539463}. It is seen mostly in the children because of insufficient aortic wall development at the place of restoration as surgery was carried out prior to the aorta has stretch out to adult size \unskip~\cite{1020278:21539462}.

\textbf{Case Report}

A male patient of 20years old was admitted in the cardiology department with the chief complaints of chest pain, SOB since 5 days. His lab values are given above in the Table~\ref{tw-9b2802a9c87e}. ECG report and Radiology report of the patient were attached below in Figure~\ref{f-357413094ca0}  and Figure~\ref{f-3a218a5afd54}  respectively. The physician suggested the balloon angioplasty with stent to the patient. On examination BP was found to be 220/100 mm Hg, Pulse rate was 80bpm on first day. On second day BP was found to be 160/70 mm Hg, PR-85bpm.On third day BP-130/70 PR-72bpm, RR-24cpm are to be found. The physician prescribed the medications like T.Nicardia retard-10mg twice a day, T.Prolomet XL-25mg once a day, T.Telvas-40 mg once a day, T.Embeta XR-50mg once a day, T.Soloposr 0.5mg once a day. Clinical laboratory values are given above as in the Table~\ref{tw-9b2802a9c87e}.


\bgroup
\fixFloatSize{images/78c1085c-d879-48aa-af5e-46206d3c5590-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/78c1085c-d879-48aa-af5e-46206d3c5590-upicture3.png}{\includegraphics{images/78c1085c-d879-48aa-af5e-46206d3c5590-upicture3.png}}{}
\makeatother 
\caption{\boldmath {ECG Report}}
\label{f-357413094ca0}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/ae679e55-c268-4afd-ac79-7882a7edd143-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/ae679e55-c268-4afd-ac79-7882a7edd143-upicture4.png}{\includegraphics{images/ae679e55-c268-4afd-ac79-7882a7edd143-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Radiology Report}}
\label{f-3a218a5afd54}
\end{figure}
\egroup

    
\section{DISCUSSION}
Coarctation of the aorta is tapering of the aorta, repeatedly occurred at the left side of subclavian artery, it also occurs at different regions of aortic arch or even at the thoracic aorta or abdominal aorta. The tapering of the aorta results an abnormal increase in the blood pressure, then it causes above extremity high blood pressure. Un-functional coarctation gives on to early CAD (coronary artery disease), LV dysfunction, thoracic aneurysm. Transcatheter treatment of Co A set off more familiar with the evolution of balloon angioplasty by Lock et al. in 1983. An unrestricted consolation of stenosis by balloon angioplasty is expertise by the destruction to the aortic inner membrane layer and media \unskip~\cite{1020278:21539465}. The considerable stumbling blocks are flinch of the vessel with relapse of stenosis was noticed after balloon angioplasty and vessel injury with resultant vessel dismemberment or aneurysm structure was noticed. Stent replacement became the primary choice of management for children and adults having native Co A and it was firstly represented by O' Laughlin in the year 1993 \unskip~\cite{1020278:21539467}. The main agenda of using foil stents was to keep away the dissection, imprudent distension and elastic flinch of aorta. It also sequel in decrease vessel injury than balloon angioplasty \unskip~\cite{1020278:21539466}. The symptoms of coarctation of aorta patients were the high or low blood pressure, enlargement of heart or murmurings, cyanosis from insufficient blood flow, declined pulses, non fulfillment to flourish, SOB. Obstacles of Co A are monotonous Co A, thoracic aneurysm, HTN, Cerebral aneurysm, heart myopathy, hemiplegia, Post coarctectomy syndrome. In this case, there are abnormalities of aorta measurements were found so balloon angioplasty with stent was advised to the patient for better outcome.
    
\section{Conclusion}
Coarctation of aorta is a long-lasting disease which sequel the whole pre-coarctation arterial base. Very close \& regular monitoring and hostile regulation of obstacles is pivotal. Among the modern progress in stent technology, one is the evolution of biodegradable stents that offers a remedy or way out for endovascular treatment for newborn and infant coarctations. Stent replacement retains the coarcted aortic portion unlock and terminate over a span of months. The framework may fabricate in most of the patients and creates a significant re-narrowing. Clinical trials are even now to manifest the practicability and effectiveness of this mechanization. But this study indicates the benefit of balloon angioplasty in the resolvation of coarctation of the aorta and also re-coarctation with the minimal morbidity. Although an observational long term assessment for hypertension and further complications are compulsory in the patients.

\textbf{Funding Support}

The authors declare that they have no funding support for this study.

\textbf{Conflict of Interest}

The authors declare that there is no conflict of interest.

\textbf{Informed Consent}

We have obtained the informed consent form from the patient to publish the case report.
    
\balance 

\bibliographystyle{unsrt}

\bibliography{\jobname}\vspace*{20pt} 
\begin{mdframed}[backgroundcolor=cpboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=1pt,rightmargin=1pt,innerleftmargin=1pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=1pt,skipbelow=1pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Copyright:}~\copyrightstmt}	
\end{mdframed}
        
\begin{mdframed}[backgroundcolor=citeboxcolor,hidealllines=true,innerbottommargin=7pt,innerrightmargin=4pt,rightmargin=4pt,innerleftmargin=2pt,leftmargin=2pt,skipabove=1pt,innertopmargin=2pt,innerbottommargin=4pt,skipbelow=2pt]%
{\fontsize{8.5}{10}\selectfont\textbf{Cite~this~article:}~\citeauthor.~\citetitle.\space\makeatletter\abbrvJournalTitle\ \@copyrightYear;\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi\ifx\@journalIssue\@empty\else(\@journalIssue)\fi:\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}.\makeatother}	
\end{mdframed}
\vskip-1.1pc

\vskip.5pc
{\includegraphics[width=5cm]{pharmasprings_logo.png}}
~\\
{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \makeatletter\@copyrightYear\makeatother\ Pharma Springs Publication.}

\end{document}
